Author's response to reviews

Title: Associations of ECP (eosinophil cationic protein)-gene polymorphisms to allergy, asthma, smoke habits and lung function in two Estonian and Swedish sub cohorts of the ECRHS II study

Authors:

Ulla-Britt U-B Jönsson (ulla-britt.jonsson@medsci.se)
Lena L Lena Douhan-Håkansson
(Lena.douhan-hakansson@akademiska.se)
Rain R Jögi (LRain.Jogi@kliinikum.ee)
Christer C Janson (christer.janson@medsci.uu.se)
Per P Venge (per.venge@medsci.uu.se)

Version: 2 Date: 13 January 2010

Author's response to reviews:

Dear Sir,

Our paper was originally submitted to Respiratory Research, but the editor informed us that it would be more suitable for Pulmonary Medicine. We agree to this and would therefore be pleased to have our paper considered for publication in your journal.

Yours Sincerely

Per Venge, MD PhD
Professor
Uppsala University